Author's response to reviews

Title: The study of expanded tri-lobed flap in a rabbit model: Possible flap model in ear reconstruction?

Authors:

Dr Eray Copcu (copcu@lycos.com)
Mesut Yazici (myazici@adu.edu.tr)
Barlas Etensel (betensel@adu.edu.tr)
Yakup Yurekli (yyurekli@adu.edu.tr)
Muharrem Balkaya (mbalkaya@adu.edu.tr)

Version: 4 Date: 11 Dec 2003

Dear Editor,

Thank you very much for your E-Mail dated December 11, 2003. We are grateful for your decision about our manuscript. Revisions were performed according to your commet and listed below. I declare that the co-author is agreed on the revision.

We hope you will find the revised manuscript satisfactory. If any further changes are necessary, we would be happy to improve.

Respectfully yours,

Eray COPCU, MD

Revisions:
1. Competing Interests were added as "none declared"
2. Author Contributorship was added.
3. Figures were re-numbered and text of manuscript was revised according to the new names of figures
4. First four figures were re-cropped according to your comments.